

Reply to Reviewers: Manuscript ID; 40319

I would like to thank the reviewers for favorably reviewing the editorial article.

Minor correction of page 2, line 10 has been corrected.

The references were checked according to the journal format.

Best wishes,

Amr El-Shazly